

A Systematic Review on Pill and Medication Dispensers from a Human-Centered Perspective

Luigi Gargioni¹ · Daniela Fogli¹ · Pietro Baroni¹

Received: 29 June 2023 / Revised: 8 November 2023 / Accepted: 7 February 2024 / Published online: 22 February 2024 © The Author(s) 2024

Abstract

As medication adherence represents a critical challenge in healthcare, pill and medication dispensers have gained increasing attention as potential solutions to promote adherence and improve patient outcomes. Following the PRISMA (Preferred Reporting Items for Systematic reviews and Meta-Analyses) methodology, we carried out a systematic literature review on papers indexed in Scopus and PubMed, which present solutions for pill or medication dispensers. Given the importance of user acceptance for these solutions, the research questions of the survey are driven by a human-centered perspective. We first provide an overview of the different solutions, classifying them according to their stage of development. We then analyze each solution considering its hardware/software architecture. Finally, we review the characteristics of user interfaces designed for interacting with pill and medication dispensers and analyze the involvement of different types of users in dispenser management. On the basis of this analysis, we draw findings and indications for future research that are aimed to provide insights to healthcare professionals, researchers, and designers who are interested in developing and using pill and medication dispensers.

Keywords Pill dispensers \cdot Medication dispensers \cdot Healthcare technology \cdot Patient-centered care

⊠ Luigi Gargioni luigi.gargioni@unibs.it

> Daniela Fogli daniela.fogli@unibs.it

Pietro Baroni pietro.baroni@unibs.it

¹ Department of Information Engineering, University of Brescia, Via Branze 38, Brescia 25123, Italy

1 Introduction

The global aging population is growing at an unprecedented rate [1], and as people age, they are more likely to develop chronic conditions such as diabetes, hypertension, and cardiovascular diseases. These conditions require ongoing management with medication therapy, which can be complex and challenging for patients. Patient non-adherence to medication therapy is a pervasive problem in healthcare, particularly among patients with chronic conditions living at home [2]. Non-adherence can occur for various reasons, such as patients forgetting to take their medication, misunder-standing dosing instructions, or intentionally choosing not to follow the prescribed regimen [3]. This can lead to poorer health outcomes, reduced treatment efficacy, and increased healthcare costs. Moreover, non-adherence can be especially concerning in home healthcare settings, as patients may not have the same level of support and supervision from healthcare professionals as they would in a hospital or clinic [4].

Scientific literature reports on a high variety of smart medication and pill dispensers aimed to provide patients with reminders for taking their medication at the correct time and to avoid errors about medicine type and dosage. Several solutions proposed in literature also envisage system architectures encompassing cloud services and mobile applications that allow caregivers and doctors to monitor patients' medication adherence and manage the smart dispensers remotely. However, such solutions are still far from being widely adopted in real contexts, since several factors should be taken into account, including the direct connection with medicine prescriptions available in the national healthcare system, the management of complex therapies through automated devices without human intervention, the recharging of dispensers, their usability for elderly people, and, last but not least, the high costs.

The purpose of this paper is to carry out a systematic review with the objective of addressing the overarching research question:

What kinds of automated dispenser solutions have been proposed in the scientific literature to help patients be adherent to prescribed care?

In particular, given the fact that the adoption and actual utility of dispensers rely critically on acceptance and correct use by patients and other involved stakeholders, like caregivers and doctors, we adopt a human-centered perspective to address the above question. A limited number of literature reviews on this topic do not take this perspective, but usually deal with technological aspects of proposed solutions or are focused on specific patients' diseases. For this reason, we aim to fill this gap in literature analysis by adopting a wider perspective that considers use context and interaction issues. Accordingly, we analyze the proposals in the literature with the main goal of identifying strengths and weaknesses that may affect acceptance and correct use by the people involved.

Following the PRISMA 2020 guidelines [5], we carried out a systematic review searching in journals, book chapters, and conferences through Scopus and PubMed, which led to the retrieval of 426 scientific papers, 56 of which were selected after the screening process and examined in depth for inclusion in this study.

The paper is organized as follows. Section 2 provides an overview of the related works; Section 3 details the methodology adopted for the systematic review; Section 4 presents the study's results and addresses the four Research Questions into which the

overarching research question has been decomposed; Section 5 delves into the study's findings, current challenges, and implications for future research; and, finally, Section 6 provides some concluding remarks.

2 Related Works

Few literature reviews have been published in the last years on the topic considered in this paper. Furthermore, they are usually focused on some specific aspects and rarely adopt a human-centered perspective.

The literature review reported in [6] does not follow a rigorous protocol for searching and selecting papers from the literature and does not present a clear research question; the authors would like to investigate the characteristics of wearable devices for measuring vital parameters and combine them with an automatic medicine dispenser; therefore, they report on solutions concerning different types of wearable sensors and medicine dispensers highlighting the most important technologies used in these solutions.

The paper [7] illustrates the systematic review carried out following a precise protocol described in a paper by the same authors [8]. The focus of this review is however pretty narrow: the authors are interested in collecting information and indications about the experience of informal carers of people with dementia using assistive technologies. Among the considered assistive technologies, there are smartphone applications, falls and motion detectors, smart gas meters, social assistive robots, and many others, including medication dispensers. Thus, on one hand, the analysis aims to acquire knowledge about carers' experience and the impact of these technologies on their quality of life; on the other hand, medication dispensers are only listed among the assistive technologies mentioned in the various papers, without going into the details of their characteristics.

A scoping review carried out in 2019 is reported in [9]. The authors selected 85 articles out of 2638 for full-text review and scrutinized them focusing on the integration of smart multidose dispensing systems (SMODS) for medication adherence, where by integration they intended the usability, functionality, and acceptability of the system. Beyond a synthetic description of the proposed solutions, the review focuses on the data related to the different studies reported in the papers about medication adherence measurement and integration assessment. From the analysis, important gaps emerged related to the lack of a standard definition of medication adherence, an inadequate consideration of caregiver involvement in the studies, the limited feedback from healthcare professionals who play an important role in monitoring patients' medication adherence, and the limited number of studies that considered the use of SMODS to manage complex therapies. The outcomes of this review are very significant from a process analysis viewpoint, but we would like to complement them with a socio-technical perspective, where both technical features of the proposed solutions in terms of hardware/software implementation and user-related issues, like management of the system on behalf of different stakeholders and interaction modalities, are considered.

A scoping review is also presented in [10], but it is focused on studies that involved people with a mental health or substance use disorder. An important inclusion criterion

was that one technological intervention had been considered in the study to improve medication adherence; however, the authors define technology in broad terms, including mobile and web apps, computer vision-based systems, interactive voice response technology, smart pill bottles, ingestible sensors, and novel biomarkers. Smart pill containers are found in 34 studies out of 127 included papers, even though they are analyzed only in quantitative terms with reference to the evaluation of medication adherence.

The paper [11] presents a literature review on IT-based reminders, namely, different types of information technologies producing text, visual, or audio messages aimed at enhancing medication adherence. Particularly, the review analyzed 48 papers out of 544, including only those publications that quantify the effects of the considered reminder on medication adherence. A taxonomy is finally proposed that classifies the reminders according to 13 characteristics divided in four categories: reminder duration (long and short), reminder type (email, SMS, sensor/device, and pillbox), disease type (chronic, acute, and other), and medication adherence improvement (medium, small, zero/negative). Finally, the authors derive from the taxonomy a comprehensive framework leading to the development of medication reminders. This review can be considered complementary to ours, since it scrutinizes and classifies reminder solutions, which are typically included as an important feature in pill and medication dispensers.

The review presented in [12] has a goal similar to ours: providing an overview of current technologies to guide the development of a novel pill dispenser. This paper considers both commercial devices and devices proposed in scientific literature. Seventeen commercial devices are described in detail, selecting eight characteristics (dosage capacity, alerts to caregivers, device portability, peripheral communication, web or mobile app availability, security, multi-user capability, and power sources). As to the literature, papers were searched using Google Scholar and IEEEXplore with the same keywords that we in turn used to define our queries in Scopus and PubMed. However, the exclusion and inclusion criteria are not explained in the review, and only 16 papers (published before 2021) are synthetically described, among which 5 papers are published before 2013, 2 are not in English, 2 are only 2-page long, thus, all of them would have been excluded from our review; then, 2 papers are not indexed in Scopus or PubMed, and 3 papers were instead not retrieved with our queries in Scopus or PubMed; and only 2 papers are included also in our review, which reports extensive information on additional 54 papers, several of them (21) published since 2021.

3 Methodology

Systematic reviews enable the creation of a comprehensive view of the scientific literature on a particular topic and an understanding of the open issues and challenges that may be related to it. In this review, we used the PRISMA 2020¹ approach, guidelines, item checklist, and flow diagram [5].

¹ PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses): http://www.prismastatement.org/

While PRISMA is mainly intended for medical and clinical research, this study targets a wider audience, including computer science, human-computer interaction, and Internet of Things (IoT). As a consequence, for the sake of better comprehension by the reader, we have decided to modify the presentation and discussion order of the PRISMA items. Additionally, not all of the items were employed. The correspondence between the sections of this paper and the PRISMA items utilized in this study can be found in Appendix.

Another key difference in our application of the PRISMA guidelines compared to their use in clinical studies is that we did not differentiate between *Report* and *Study*, opting instead to solely employ the term *Paper*. According to PRISMA, *Reports* denote the actual scientific papers, while *Studies* refer to the research studies that led to specific results. As a result, PRISMA distinguishes between the research work as a whole and the publication of the results, as the results of a single study are frequently published in multiple papers. However, in our case, all 56 papers included in this review relate to distinct studies, obviating the need to differentiate between *Report* and *Study*.

3.1 Research Questions

As outlined in Section 1, the central focus of this systematic review is to respond to an overarching research question: *What kinds of automated dispenser solutions have been proposed in the scientific literature to help patients be adherent to prescribed care?*

To guide our study, we defined four specific research questions:

- [RQ1]: What stage of development is the proposed solution at?
- [RQ2]: Which are the hardware and software technologies adopted in the proposed solution?
- [RQ3]: Which are the interaction modalities provided by the proposed solutions?
- [RQ4]: What kind of user involvement is envisaged in the proposed solution?

3.2 Search Process

To include as much results as possible in our literature review, we used Scopus and PubMed. The former, which we adopted as primary source, is an online citations and abstract database.² In the comparison with other digital libraries presented in [13], Scopus turned out to provide the maximum coverage. In particular, Scopus has an extended database, with tens of millions of records from tens of thousands of journals, conference proceedings, and book series. This breadth of coverage ensures that researchers have access to a diverse array of scholarly content from different fields and disciplines, which is essential when conducting a thorough and comprehensive review. About PubMed,³ it is a vast database comprising millions of biomedical literature citations from journals, books, and online sources. It covers a wide range of topics including medicine, nursing, dentistry, veterinary medicine, and health care systems.

² Scopus - Document search: https://www.scopus.com/

³ PubMed - National Library of Medicine: https://pubmed.ncbi.nlm.nih.gov/

This is specific to the medical field, but to ensure relevant content is not omitted, research was also conducted on it.

We performed a query on Scopus on February 6th 2023 searching "*Pill Dispenser*" *OR* "*Medication Dispenser*" on the title, abstract and keywords. The only filter used was the English language. Following a suggestion provided by an anonymous reviewer, we checked our assumption about the coverage of Scopus by performing the same query on PubMed on October 23rd, 2023. The outcome confirmed our first assumption, since no further relevant papers emerged through this query, as it will be better explained in Section 4.

As for the time range, we chose the last 10 years, i.e., starting from 2013. This decision was based on the fact that the strong evolution of technologies quickly makes solutions in this field obsolete.

3.3 Paper Selection

The title, authors, year, publication venue, and abstract of each retrieved paper were extracted and organized in a collaborative spreadsheet.

As to conference papers, we included full papers, short papers, papers presented at workshops, poster papers, late-breaking work papers, demo papers, and work-inprogress papers. We omitted workshop description papers, conference description papers, and doctoral consortium papers. We also excluded literature reviews so as not to report duplicate information already in individual papers. The discussion of literature review papers is however included in Section 2.

Table 1 displays a comprehensive account of the inclusion and exclusion criteria.

Two reviewers conducted an independent screening of the papers to assess their eligibility for inclusion based on the predefined criteria. In case of discrepancies, the reviewers resolved them through discussion. Papers that did not meet the inclusion

Table 1	The inclusion a	and exclusion	criteria	employed	in this	systematic	review
---------	-----------------	---------------	----------	----------	---------	------------	--------

Inclusion criteria

IC1: Publication date between 2013 and Feb. 6th, 2023

IC2: English language

IC3: Conference papers types: full papers, short papers, poster papers, papers presented at workshops, late-breaking work papers, demo papers, work-in-progress papers

IC4: Focus on a pill or medication automated dispenser and on the relevant technological aspects

Exclusion criteria

EC1: Conference papers types: conference description papers, workshop description papers, and doctoral consortium papers

EC2: Literature reviews

EC3: Low number of pages (less or equal to four pages)



Fig. 1 PRISMA flow diagram

criteria were excluded, and the reason for their exclusion was documented in the shared spreadsheet.

A first screening step was performed by analyzing only the title and abstract of each document. Any discrepancies were discussed and resolved by mutual agreement.

Afterwards, the full texts of potentially relevant papers were obtained, and both reviewers screened them. The same approach as above was also used to resolve discrepancies in this phase. Inter-rater reliability was calculated for the full-text screening stage using *Cohen's kappa* [14].

3.4 Data Extraction

Additional columns were added to the spreadsheet to document all relevant data related to the research questions extracted from the included papers. Both reviewers thoroughly read and completed the spreadsheet independently. Afterwards, the

Table 2Papers included in thestudy classified according to thepublication type	Type of publication	Number
	Conference paper	41
	Journal article	12
	Book capter	3



Fig. 2 The number of papers published each year

two spreadsheets were compared, and any discrepancies were addressed through discussion until a consensus was reached by finally creating a shared spreadsheet.

4 Results

The PRISMA flow diagram is reported in Fig. 1.

Out of the 426 papers initially identified (394 on Scopus and 32 on PubMed), 34 were duplicate papers, and 17 papers were excluded before the screening phase because they were descriptions of conference or workshop, proceedings, or literature reviews (see IC3, EC1, and EC2).

Of the 375 papers left, 251 papers were excluded after analyzing their abstract because they were considered out of scope (see IC4). Four papers could not be retrieved/downloaded.

The eligibility of the remaining 120 papers was assessed by reading the whole content, and 64 of them were excluded because they did not meet the inclusion criterion IC4 or met the exclusion criterion EC3. In this phase, *Cohen's kappa* before reconciliation was 0.78. Table 2 shows the distribution of the selected 56 papers according to the type of publication, i.e., conference paper, journal article, or book chapter.

Figure 2 shows the number of selected papers published each year from 2013 to the date of the query performed on Scopus (February 6th, 2023).⁴We may observe how the majority of papers (42 out of 56) have been published in the last 4 years.

For the sake of brevity, in the following sections, we will use just the word "dispenser" to denote "pill dispenser" or "medication dispenser."

⁴ As already mentioned, the query performed on PubMed did not provide additional papers.

4.1 RQ1: What Stage of Development is the Proposed Solution at?

As to the stage of development of dispensers discussed in literature (RQ1), we decided to classify the papers according to three main categories: (1) those presenting a design study of the dispenser or a very preliminary prototype; (2) papers illustrating the design, development, and (in some cases) evaluation of a working prototype; (3) papers discussing survey studies or comparative studies of commercial systems.

In the following, we provide a brief description of all the 56 selected papers, classifying them according to the above categories (see a summary in Table 3). To answer the other research questions, we will consider only the papers that provide the most significant information.

4.1.1 Design Studies and Preliminary Prototypes

Most of selected papers (35 out of 56) present a design study or a very preliminary prototype.

The work in [15] proposes the design of Medi-Deep, a comprehensive solution to address non-adherence problems. The architecture of Medi-Deep is outlined in the paper as a centralized system consisting of four components: the health record system (HRS) shared by hospitals, clinics, and pharmacies; the patient information system (PIS) that allows physicians to monitor if and what medicines are taken by a patient; the intelligent drug container (IDC) that is the hardware component at patient's home; and the mobile tracking system (MTS), namely, a mobile application to provide patients with reminders and various types of information medications.

In [16], a comprehensive solution is delineated as well, which is specifically focused on patients with mild or moderate Alzheimer's disease. Therefore, it includes as hardware components a simple dispenser and a set of alarm panels to be installed in house rooms to provide reminders through several alarm types. Different software components for the nurse/caregiver, the drugstore, the physician, and the hospital, all interacting with one another, are then hypothesized in the solution. Only preliminary prototypes of the medication dispenser and the application devoted to the nurse/caregiver are briefly presented in the paper.

A survey study and a design study are presented in [17]. As to the former study, state-of-the-art dispensers were first analyzed; then, 38 elderly people were requested to fill in an online questionnaire about their medication intake; finally, the usability of three commercial dispensers was qualitatively evaluated with 17 participants. For the latter study, 16 design variations of a modular dispenser concept were created and

Stage of development	Total	Papers
Design studies and preliminary prototypes	35	[15–49]
Working prototypes of pill dispensers	13	[50-62]
Studies of commercial products	8	[63–70]

Table 3 Summary for Section 4.1

evaluated by experts of medical products. Three design concepts reached high consent, but their evaluation regarding the user interaction with the dispenser is left for future work, while functional and system requirements of an IoT-enabled dispenser and a preliminary hardware prototype are presented by the same authors in [18].

The solution proposed in [19] aims to overcome the issues of commercial dispensers existing at the time of paper writing (2019) concerning mobility, medication management, monitoring, updating medication schedule remotely, and authentication. It is an envisioned mobile dispenser designed for smartphones; it consists of software modules responsible for controlling and monitoring the dispenser, with data stored locally and on the cloud, and of a hardware module shaped as a mobile case with three cabinets locked with magnetic locks; the cabinets are opened only when attached to the smartphone, and there is a command to take or insert medicine.

A prototype of an IoT-based model of a multi-user dispenser using fingerprint authentication is presented in [20]; the paper reports the flowchart of the dispenser operation and a brief description of the developed hardware components.

In [21] and [22], proofs-of-concept medication dispensers are described by delineating their operation and the blocks composing the overall system; preliminary hardware prototypes with very limited functionalities are then presented. Similarly, [23] and [24] present the electronic components of basic and low-cost dispensers. The paper [25] describes the hardware architecture and operation of a medicine reminder and dispensing machine designed for people living in Indian local areas, who can only speech their local language; thus, the system includes an audio module recording reminders in the desired language.

The prototypes presented in [26] and [27] are preliminary as well, but, differently from the previous ones, they have been designed to send SMS messages to alert the patient when it is time to take medications; also, in these cases, a brief description of the components is provided, along with photographs showing their assembly. The low-cost characteristic of the proposed dispenser is highlighted in [28] and [29]: the former describes a crude implementation of a prototype, realized with local materials (wood and plastic), intended to be used in multi-person homes in developing countries; the latter illustrates how CAD software and 3D printing can be exploited to build a low-cost, strong, and light structure of the dispenser, even though its peculiar aspect is a mechanism to count the pills to be dispensed even in case of different pill dimensions. The designed dispensers are intended to be easier to use and cheaper than commercial ones, but no real experimentation is discussed. With a similar goal, the development of an automated dispenser for persons with neurodegenerative diseases living in low resource settings is presented in [30]; this simple and inexpensive device can automate counting and dispensing of medicines of different sizes with a pretty high accuracy, according to the system trials reported in the paper.

The dispenser described in [31], instead, is capable of automatically sorting pills within a container using algorithms for image recognition and optical character recognition (OCR); the work is however mostly focused on the evaluation of precision and accuracy of the algorithms, while it gives a brief account of the hardware and of the management application running on a personal computer. Image processing and OCR algorithms are also the main focus of [32], where the goal in this case is scanning prescriptions and then guiding the patient in filling the automatic dispenser; the device

provides patients with reminders to take medications and doctors with information about patients' medicine intake.

In [33], the block diagram and hardware components of a dispenser working as a reminder system for patients are described; its main novelty with respect to other proposed preliminary prototypes consists of a cooling system and storage compartments also for vaccines. Similarly, the hardware components of a simple dispenser are illustrated in [34, 35]; both papers also include a brief description of the Android application to be used to set the time slots for medicine schedule and keep track of pill taking. A mobile application is also part of the IoT-based medication management system that completes the dispenser proposed in [36]. In this case, the dispenser should serve different old residents in a center, and caregivers must use the app to manage medication treatments and receive notifications if patients do not take a medicine or the dispenser needs to be refilled. The proposal illustrated in [37] is focused on the cloudbased architecture of the dispenser, which allows patients, doctors, and pharmacists to monitor data about medicine intake and receive notifications.

Paper [38] reports on a smart dispenser combined with a pulse measuring system. The hardware prototype is delineated and shown with photographs, as well as the software component that consists of a web application for dispenser configuration and monitoring patients' pill intakes. The dispenser presented in [39] includes a heartbeat and a temperature sensor; this device is simple, easy to use, and low cost, but its features are limited.

The dispenser proposed in [40] is meant to not only remind users to take medicine at the right time, but also to provide the correct pills and doses. A preliminary version of the system, developed using 3D printing and Arduino microprocessors, is briefly illustrated in the paper. Paper [41] investigates the role of IoT technology in healthcare applications, services, and products; then, the paper proposes a model of an IoTenabled dispenser that is meant to be so simple and easy to handle that every elderly patient can use it easily; a cloud database accessible by the doctor would play an important role to store information about medication therapy and patients' pill intake. The hardware components of a prototype dispenser are discussed in [42]; the original contribution of this work is the proposal of integrating this device in a smart wheelchair, but no testing or evaluation with users is presented in the paper. In [43], a syringe dispenser for diabetic patients is combined with a pill dispenser; the block diagram and a very preliminary realization of the system are then presented.

A modular dispenser is proposed in [44]; modularity is achieved by the number of boxes that contain the pills, which may vary according to the patient's therapy and possible therapy changes. The solution is paired with a mobile application that allows therapy configuration and management and pill dispenser monitoring. A pilot study performed with five participants allowed to gather feedback on how to improve the mobile application.

In [45], the authors propose an automatic dispenser that encompasses a portable device and a fixed device that can communicate with each other through a USB cable; in particular, the portable device can retrieve the time and dose data of the medical treatment from the fixed device. The results of a 2-week experiment carried out with one female subject show how the subject improved her habit to follow the medical treatment. Notwithstanding these good results, the solution can be regarded as a proof

of concept, since it requires several improvements in terms of performance, aesthetics, and integration with smartphones.

The preliminary prototype described in [46] is a moving dispenser that, beyond having features similar to those ones of other proposals in terms of hardware components and software application (i.e., a mobile app to set the time for medications), is able to deliver pills to the patient's location, thanks to the implemented simultaneous localization and mapping (SLAM) algorithm. In [47], a different solution to medication adherence is proposed: the idea is designing and developing a mechanoid social robot able to detect the location of the patient through sensors (e.g., beacons) and dispense them the medicine into a container, by prompting a text message to the caregiver. To assess the feasibility of the idea, a lab-scaled prototype was developed using the LEGO Mindstorm toolkit. Performance tests were carried out on the prototype in different environments, and a user acceptance test was performed through an interview after explaining the technology. An autonomous robot to dispense medicines at the patient's location is proposed also in [48]; it uses a mobile application to set the therapy, receive alert messages, and manage the interaction with the robot. The idea has been however only simulated at a software level. Similarly, [49] proposes a medicine dispenser robot for hospital wards and retirement homes, which can also carry out patient monitoring and voice assistance with prerecorded audio traces; a comprehensive solution is thus delineated, which includes an app for therapy management, a camera for patient surveillance and video recording, a temperature sensor, a speaker, and a microphone to acquire patient's request.

4.1.2 Working Prototypes of Pill Dispensers

Thirteen out of 56 papers describe complete and working prototypes of dispensers that have been tested in terms of functionality and/or usability.

One of the first and almost complete research works is that presented in [50], where the authors adopted an iterative prototyping approach including phases of requirements analysis and concept design, prototype development, and user evaluation on the field. The personal medicine assistant is not properly a medicine dispenser, but a computerbased multi-modal reminder system that shows a medication plan and reminds the patients about the medicines to take with audio-visual information, and allows health professionals to plan the therapy and monitor medicine adherence. Two field trials, with four elderly people each and lasting 2 and 3 weeks respectively, provided satisfactory results about usefulness and user experience, as long as interesting feedback to improve the system.

The paper [51] illustrates the DIALCAT platform; it encompasses an IoT-enabled dispenser and a mobile application to obtain tailored information and communicate with doctors, caregivers, and pharmacists. The mobile application can also be used by caregivers and doctors to monitor the patient's health and medication adherence and by pharmacists to receive orders to replenish the dispenser. At the time of paper writing (2019), a controlled clinical trial with 58 hospital patients in the control condition, 58 patients using the DIALCAT platform, and 58 patients who, beyond using DIALCAT, should have received periodic feedback on their metabolic control was planned.

In [52], the low-level details of the hardware implementation of a dispenser are discussed; then, a brief description of the software application where the user can insert the required time and the intake interval of each medication, and set related alerts, is provided. The paper also mentions a trial to assess the usability of the dispenser and its acceptability by patients, claiming that it helped geriatric patients remember to take their medications, thanks to its intuitive user interface; however, no specific information about this trial is provided in the paper.

The paper [53] describes a dispenser that, although very simple, can successfully remind patients to take their medicines via SMS and dispense pills of various sizes and shapes. The dispenser has been given to ten patients for a long-term test; after 10 weeks, a survey was conducted to collect information about the user experience, which resulted to be very satisfying, also because the system helped participants increase their medication adherence.

The authors of [54] explain how they extended the preliminary prototype presented in [28] to support patient-doctor remote communication, and report on a quality study based on key informant interviews performed with 17 health workers in four hospitals in Uganda; most of the participants considered the dispenser useful and easy to use, but they also suggested several improvements before actual deployment with patients could take place.

The work presented in [55] aims to develop a system that can automatically dispense pills of various sizes and shapes. The system, called PharmAid, consists of a portable wearable device providing reminders to users to take their medications and of a dispenser storing a month's supply of pills. The inner pill container is able to move along the x, y, and z axes, and Bresenham's algorithm is used to control the motors enabling the 3D movements; the paper is focused on its implementation in PharmAid and on the realization, assembly, and interaction of all the electronic components of the device. Simulation and stress tests contributed to study an improved version of the system that reduces dispensing time.

The paper [56] explains the design methodology and the hardware/software realization of a simple dispenser that rotates every 6 h making a pill drop in the container and triggering a buzzer to alert the elderly patient that it is time for the pill. Using Wi-Fi, the system also sends notifications to the caregiver and records the medication drop data for further analyses. The paper mentions user testing, without providing details about participants, and discusses possible improvements after tests.

The system proposed in [57] aims to combine IoT hardware and software technologies to create a system that encompasses a dispenser, a back-end software infrastructure, and a mobile application. The dispenser is modular and consists of five or more slots and can manage multiple users. A complete hardware and software system is proposed; a comparison with existing commercial dispensers shows its superiority in terms of costs and features. However, a usability and user experience evaluation has not been carried out. The same limitation affects the proposal presented in [58], where only functionality tests have been performed on the hardware device and mobile application developed. Interestingly enough, in this case, the mobile application is not only used to set up the therapy and monitor medicine intake, but it provides the patient with an easy way to operate the dispenser and obtain information from it, such as the number of pills left in the dispenser. A similar approach is followed in [59], where a dispenser for multiple users is presented; the paper mainly describes the mobile application used to set therapies, manage the dispenser, and send notifications to caregivers; also, in this case, tests concerned only system functionalities and not usability and user acceptability.

User tests with three participants have instead been performed on the smart medication planner (SMP) described in [60]. SMP is intended to help blind people manage their medication; thus, its architecture is studied to satisfy blind people's needs, especially interaction based on voice communication. Even though SMP is still a prototype, its cost, portability, and usability resulted to be quite satisfactory for the users; however, some weaknesses were recognized such as the fact that voice recognition required a loud and clear voice and that the device packaging and size needed some improvements.

The solution proposed in [61] aims to cope with the problem of determining if the patient actually had ingested the medication taken from the dispenser; this is particularly important for patients with dementia. The authors propose a dispenser able to monitor the patient through sensors and to vocally guide them to complete the medication intake correctly. System evaluation (particularly activity recognition) was performed in a laboratory with 20 participants (faculty members and students aged 20 to 60) showing high efficacy of the proposed system; however, participants did not have dementia nor they suffered of health problems. The dispenser presented in [62] has been designed also considering patients who are paralyzed, deaf or blind, or have mental diseases, and is intended to be deployed in developing countries due to its low cost. The dispenser was tested by a female patient for 14 days at home; she had heart disease and a prescription to take medicines three times a day. Before the use of the dispenser, she forgot to take medicines at least 25% of the time, while, from data collected automatically by the system, it was observed that the forgotten rate decreased to 11.90%, with a significant difference between the first and the second week of dispenser usage (in the second week, the patient forgot only once to take medicines). Notwithstanding these good results, the experimentation with only one user is too limited to derive useful hints.

4.1.3 Studies of Commercial Products

Eight papers out of 56 present studies about commercial pill dispensers.

The paper [63] describes the evaluation in terms of acceptability and technical robustness of the Elucid Pill Connect system, a smart pill bottle dispenser. The use of the dispenser requires to interact with a dedicated mobile application. The study involved ten healthy participants that used the system with placebo capsules for 14 days. At the end, they completed the System Usability Scale (SUS) questionnaire, responded to additional satisfaction questions, and filled in the Medication Adherence Report Scale (MARS); in addition, they participated in a short interview to report problems that occurred during the trial and suggestions to improve the system. From the results, it emerged that, with this system, medication adherence was very high and participants judged the system acceptable from a usability viewpoint. Suggestions for

system improvement include reminder personalization and additional features in the mobile app for training and accessing medication intake information. Finally, a couple of technical issues were found in the bottles to be considered by the manufacturer. The fact that the participants involved in the studied did not use prescribed medication, but volunteered to take placebo capsules, is the major limitation of this work.

The paper [64] describes an ethnographic study about the use of a robotic medication-dispensing system in the sparsely populated areas of Finnish Lapland. The system is an assistive technology reminding when medicines should be taken, which provides companion-like features. It was already on the market before the study took place. The study aimed at investigating how elderly people domesticate the system in their everyday lives and which meanings caregivers and health professionals assign to the robot during the domestication. The concept of *domestication* was used in the study, instead of adoption or use of technology, since it encompasses the meanings and values people assign to a given technology, how the technology is experienced, and which roles it plays in people's everyday lives. The system dispenses tablet medicines on a regular basis; it can be remotely monitored and can be used for message exchanges between patients and the home-care service professionals. Eleven participants were involved in the study, including five elderly patients (73-89 years old), four nurses, and two other healthcare professionals. All the five patients lived alone and had impaired memory. Data from semistructured interviews and observations (with photographs) were collected along a 5-month period. One interesting outcome was that the patients considered the system as easy to use, but they could not use it without the nurses filling it with medicines regularly. Support for learning the system was also needed, but at the end, all patients were satisfied with the system, also thanks to the help of their social network (nurses and children). Some concerns about the system emerged related to its aesthetics and its inflexibility in dispensing medicines at specific times.

The paper [65] analyzes four commercial products—MedaCube, MedMinder, Hero, and Pria—that can dispense medicines automatically at the time predefined by the user (usually, a caregiver when the patient is an elderly person). Then, it presents a comparison based on design, pill refilling, services provided, sound and light alerts, security, and supervision features. Finally, an interdisciplinary team studied the user experiences in different settings (home environment, hospital environment, and nursing home) and arrived at defining the business model of a product, whose requirements, overall architecture, and basic device representation are described in [66].

In [67], the focus is on the evaluation of usability and task workload of dispensers, underlining how these aspects are critical to medication adherence of elderly adults. The paper reports on the quantitative findings (SUS (system usability scale) scores and NASA-TLX (NASA task load index)) of a study carried out in Canada, which examines 21 commercial products with the participation of 23 elderly adults (65 years old or older), five caregivers, and 11 health care professionals. Each participant tested five electronic medication adherence products, executing a series of tasks according to a mock polypharmacy medication regimen designed for the purpose of the study. The study demonstrates that some dispensers may be easier to use than others; however, the study has been carried out in an unfamiliar environment, without training,

and with the request of using five different systems; this probably increased participants' anxiety, frustration, and task load during the test. No design implication for the development of novel dispensers is provided in the paper. To overcome some of these limitations, in [68], the same authors present the results of a study about the use of the *Spenser* commercial dispenser at patients' homes for a 6-month period. In this case, 58 patients and 11 caregivers were recruited, and usability, usefulness, and satisfaction were assessed through the SUS and the usefulness, satisfaction, and ease of use (USE) questionnaires, beyond medication adherence, impact on caregivers, and capability of pharmacists to identify problems. The results revealed a high mean medication adherence rate of (98%), and a very high SUS score (85.74); in addition, the dispenser allowed pharmacists to conduct medication reviews and identify drug therapy problems. Even though design implications are not reported here either, the study provides information to design future testing studies of similar products.

The paper [69] presents a randomized controlled study to compare the efficacy of Medido Connected, a commercial dispenser, versus regular care, in patients with Parkinson's disease. Data were obtained in the Medido group and control group at baseline (36 vs 51 participants), after 3 months (24 vs 36 participants), and after 6 months (29 vs 45 participants) follow-up, by means of validated questionnaires assessing physical disability, quality of life, non-motor complications, and caregivers' quality of life. A clinical improvement of physical disability in the Medido group was observed for the elderly and more severe patients. However, no significant differences were observed for the other patients. The authors hypothesize a negative influence on results of some limitations of the study, such as the many missing measurements at 3 months of follow-up (due to illness of the senior researcher), and the possible Hawthorne effect on the control group (improvements in therapy adherence may be due just to the inclusion in the study).

Finally, the paper [70] investigates the deployment of a medication management robot system (MMR) in home healthcare services of a municipality in Sweden, to understand the impact it could have not only on the elderly community, but also on health sector policymakers, elderly care managers, care personnel, and family members. The MMR involves a medicine dispenser that can administrate machinepacked dose sachets and a telecare system. The preparation of multidose sachets is carried out by the pharmacist who may access the system where the physician enters prescription information. Medicine sachets are delivered to a nurse; the nurse or their delegated home care person inserts the medicine sachets into the medicine dispenser and enters the medication information in the MMR software. The telecare system is used by the nurse and can monitor the MMR to alert caregivers if needed. A qualitative research approach (interviews, focus groups, workshops, and observations) is adopted in this study. Participants appreciated the system, which was perceived as easy to use, safe, and able to save time and costs for nurses dispensing medications themselves for patients. However, several interesting issues emerged from the study related to the organizational and inter-organizational complexity and the importance of information and communication quality. The paper describes these issues in detail and proposes further investigations on the topic.

4.2 RQ2: Which are the Hardware and Software Technologies Adopted in the Proposed Solution?

In order to address RQ2, an examination was conducted to discern the hardware and software elements outlined within the various proposed solutions documented in the analyzed papers, with the aim of understanding their construction. To facilitate clarity and organization, this section has been subdivided into distinct categories, specifically *Electronics, Mechanics*, and *Software*, wherein the different components have been appropriately grouped.

4.2.1 Electronics

The main classes of electronic components we identified are microcontrollers, communication-related components, sensors, the real-time clock, and storage devices (see a summary in Table 4). Please note that this table and the following ones only include references to papers that contain information related to the table subject; papers where not relevant information is provided are omitted.

Microcontrollers Microcontrollers, such as Arduino, Raspberry Pi, and NodeMCU, are popular platforms used in the field of electronics and embedded systems. They provide a wide range of features and capabilities and offer different levels of computational power, flexibility, and connectivity options. The choice of the platform depends on the specific project requirements, complexity, and personal preferences of the developer. These platforms have vibrant communities and extensive online resources, making them accessible to both beginners and experienced developers alike.

A microcontroller can be used in a pill dispenser to control the dispensing mechanism and monitor the usage of the medication.

Electronic component	Total	Papers
Arduino	29	[16, 20–22, 25–27, 29–36, 38–40, 43–46, 48, 49, 56, 57, 59–61]
Raspberry Pi	10	[18, 20, 32, 34, 37, 41, 52, 57, 60, 66]
NodeMCU	2	[49, 62]
GSM	6	[26–28, 36, 53, 54]
Wi-Fi	14	[25, 34, 35, 39, 41, 44, 48, 49, 52, 53, 55, 56, 58, 60]
Bluetooth	9	[18, 22, 33, 40, 43, 46, 55, 59, 60]
NFC	2	[16, 50]
IR	11	[18, 20, 25, 27, 34, 36, 47–49, 51, 53]
Camera	6	[16, 31, 32, 40, 49, 66]
Kinect	1	[61]
Fingerprint	3	[20, 32, 33]
Real-time clock	4	[21, 23, 27, 53]
Storage	4	[21, 25, 44, 49]

 Table 4
 Summary for Section 4.2.1

261

Arduino is used in most of the prototypes presented in the different papers. This is explicitly described in 29 papers, but the number could also be higher as in some papers no special attention was given to describing what hardware devices were used. Different Arduino models are used according to the computational power and compatibility requirements. The models used, where specified, are the *Mega*, *Uno*, and *Nano*. Using an Arduino in a dispenser prototype provides a flexible and cost-effective solution for designing a customized medication management system that can be easily adapted to meet the specific needs of different patients. In [20] and [25], the use of the Arduino within the proposed prototype is described in detail.

Raspberry Pi was used in ten prototypes. The hardware components in a dispenser, such as motors and sensors, can be connected to the Raspberry Pi through its GPIO pins. The Raspberry Pi can then be programmed to control the dispensing mechanism based on a schedule, triggered by sensors or user input. In addition to hardware control, the Raspberry Pi can provide a user-friendly interface for setting up medication schedules, monitoring dispensing events, and receiving alerts for missed doses. This interface can be connected to a display and input/output devices such as a keypad or touchscreen. Furthermore, the Raspberry Pi can store data on medication usage and potentially be integrated with other health monitoring systems such as wearables or health apps. This data can be analyzed to provide insights on medication adherence and potential health risks. The Raspberry Pi offers a versatile and cost-effective platform for building a customizable dispenser prototype with advanced data management capabilities. The use of Raspberry Pi is explained for example in [57] and [52].

Only two papers present NodeMCU as the main microcontroller. NodeMCU is a development board based on the ESP8266 microcontroller, designed for IoT. It offers an integrated Wi-Fi connection and a 32-bit processor and supports for the Lua programming language. In a dispenser, NodeMCU can be used to connect the device to a Wi-Fi network and send notifications to the patient or caregiver about the time of medication administration. In addition, it can be used to monitor the medication level inside the dispenser and to send alerts when a refill is needed. Wi-Fi connectivity also allows the dispenser's data to be synchronized with a mobile application or web server, so that it can track a patient's adherence to drug therapy and send reminders and alerts in the event of any problems or errors in medication administration. In [49], NodeCMU is used for a prototype of a dispenser equipped on board of a mobile robot, while in [62], it is used in a standard prototype as the main processor of the device.

Communication Components Communication components play a vital role as they enable devices to transmit and receive data without a wired connection. Each technology has its own strengths and applications, and the choice of communication technology depends on factors such as range, power consumption, data rate, and the specific requirements of the project or application.

In six papers, a GSM module is included in the implementation. A GSM module is incorporated into a dispenser prototype to enable communication between the dispenser and a user or caregiver's mobile device. Once the GSM module is connected, the pill dispenser can send text messages or make phone calls to a designated mobile number. This allows for remote monitoring and management of medication schedules, as well as alert notifications for missed doses or low medication levels. The use of a GSM module is central in [53] and [54].

The use of Wi-Fi is reported in the prototypes described in 14 papers. Wi-Fi can be used to connect the device to other applications or online services, allowing remote access and monitoring of the dispenser by caregivers or clinicians. An example of the use of Wi-Fi is presented in [56] and [49].

The use of Bluetooth is reported in nine papers. It is used to connect with a smartphone or tablet, which provides access to pill dosage information and the ability to change dispenser settings. In [43], this technology is used to make the main Arduino Uno microcontroller communicate with the Blynk IoT Platform for the exchange of the necessary information to manage the therapy. In the system presented in [46], it is used to connect the management app to the dispenser. In [55], Bluetooth creates a master-slave communication from the main device to the wearable one. The system proposed in [60] exploits Bluetooth connections to link the smart medicine planner with the smart medicine box. Furthermore, when the user requests to refill the medicine box, the program verifies the Bluetooth connection with it in order to dispense the medicine. In [40], the main Arduino controller receives the schedule of medication timing and type settings from a smartphone app through a Bluetooth wireless connection. The authors of [18] demonstrate the use of Bluetooth interfaces as a health gateway for acquiring vital signs; the dispenser is also equipped with alarm capabilities that communicate with the app. The exchanged data include the times when the box is opened and the reminder settings. The purpose of the study presented in [22] was to connect Bluetooth with an LCD display to facilitate the counting of pills in a tray, which could then be conveniently displayed on mobile phones or personal computers. Finally, in [59] and [33], the Bluetooth connection between the smartphone and the Arduino microcontroller allows the smartphone to transmit commands indicating which container and motor to activate, in order to dispense a specific pill from the container.

NFC technology is used in [50] and [16]. In the former, it is used to verify the medicine package, while in the latter, it is used to connect to the smartphone application of the nurse.

Sensors In 11 papers, an infrared (IR) sensor is mentioned. The IR sensor is a hardware component that detects the presence of objects using infrared technology. In a dispenser, the IR sensor can be used to detect the presence of the user's hand, so that the dispenser motor is only activated when the user is ready to receive pills or also to detect the presence of the pill in the compartment. For example, in [27, 51] and [36], this type of sensor is described.

In six included papers, a camera is used in the prototype implementation. In a dispenser, a camera can be used to verify that the patient has actually taken the pill, to recognize pills or prescriptions through image processing, or to create a direct communication with the patient. For instance, in [40], the proposed system incorporated a camera for capturing images of pills, which are subsequently transmitted to the app for recognition using convolutional neural networks. The solution presented in [31] uses the camera for OCR and medicine sorting, while in [32], it is used for scanning In [61], an approach for monitoring medical adherence via Kinect is presented. The Kinect is a motion sensor developed by Microsoft for the Xbox console. This technology uses a combination of cameras and depth sensors to track users' movement and detect the position of objects in their surroundings. In this prototype, the Kinect is used to track the user's movement while taking their medication.

In [20, 33] and [32], a fingerprint sensor is included in the described prototype. In a dispenser, it provides secure access control by using unique fingerprint patterns for user authentication. It enhances security and allows for monitoring medication usage and adherence.

Real-Time Clock The real-time clock is a hardware component that keeps track of time precisely and continuously. In a dispenser, the real-time clock is used to keep track of time and date so that it knows when to dispense pills. The real-time clock can be programmed to activate the dispenser motor at specific times, ensuring that pills are dispensed at the right time. In addition, the real-time clock can be used to keep track of the last time the pills were dispensed, helping to prevent overdosing or underdosing. We assume that a real-time clock is included in all the prototypes presented, even when not explicitly mentioned. The description of this component is explicitly provided in four papers: [21, 27, 53] and [23].

Storage In four papers, a secure digital (SD) card is used in the described implementation. In a pill dispenser, an SD card can be used to save pill dispensing data, allowing easy access and retrieval of information. The use of an SD card for the purposes just described can be found in particular in [44] and [21].

4.2.2 Mechanics

The mechanical components play a crucial role in enabling motion and physical interaction. The choice of mechanical components depends on the specific requirements of the application, including motion requirements, load capacity, precision, and environmental factors. In particular, several kinds of motors are used in the surveyed solutions.

A stepper motor is a type of electric motor that moves in a precise and predefined manner, step by step, when powered. These motors are often used in automation and robotics devices, where high precision is required. In a dispenser, the stepper motor is used to dispense the correct amount of pills. The motor is programmed to move the dispenser by a certain number of steps to dispense the desired dose. This is explicitly described in eight papers, for example, in [35] and [46].

In [23], a brush motor is used. It is a type of electric motor that uses brushes to supply power to the moving parts of the motor. These motors are typically used in small devices.

A servo motor is used in six prototypes. It uses a feedback system to maintain a specific position. These motors are often used in devices that require precise position control, such as drones or robots. The use of a servo motor is described for example in [43] and [48].

The motor driver is a hardware component that manages and controls the motor movements. It is a fundamental component, so most of papers do not describe it. In [42, 53] and [39], it is however explicitly mentioned.

4.2.3 Software

This section presents an overview of the software components adopted in the examined dispensers distinguishing software at the application level and at the infrastructure and system level (see a summary in Table 5).

Application Level At the application level, a mobile or web application is usually developed to manage and automate the process of medication dispensing. The application is designed to ensure medication adherence, organize medication schedules, and provide reminders to users. In addition, a cloud-based solution is often proposed to guarantee the storage and retrieval of information by different types of users (patients, caregivers, doctors, etc.). The application level for a dispenser focuses on creating a usable, reliable, and effective solution.

In 22 of the analyzed papers, an Android or web-based application is developed. An Android app is a software application designed to run on Android mobile devices. In a pill dispenser, an Android app can be used to monitor the status of the dispenser, display medication administration data, and send notifications to the patient or caregiver regarding the time of medication administration. In addition, the app can be used to synchronize the dispenser's data with a cloud server or other applications. In several papers, it is not specified whether it is an Android app or whether it is a web-based application, but the functionalities provided by the two technologies are quite equivalent. Some papers describing the use of these platforms are [33, 34] and [39].

Software component	Total	Papers
Android/web-based application	22	[15, 16, 18, 33–41, 44, 46, 48, 49, 51, 52, 57–59, 63]
Blynk IoT platform	2	[43, 56]
ThingSpeak	1	[56]
Cloud server	6	[18, 25, 37, 42, 51, 58]
MQTT server	2	[52, 58]
ROS	1	[46]

 Table 5
 Summary for Section 4.2.3

In [56] and [43], the Blynk App for smart homes is used. Blynk IoT platform is a platform that makes it easy to create apps for IoT devices and control them remotely. Blynk provides open-source code libraries and an intuitive user interface that allows app creators to drag and drop graphical elements to create customized interfaces. In the dispenser context, Blynk could be used to create a mobile app that allows users to monitor their medication dosage, set reminders, and keep track of their medication intake. In addition, in [56], the ThingSpeak service is used to record the medication drop data. ThingSpeak is an IoT service for collecting, analyzing, and visualizing data from sensors and devices connected to the Internet.

In six papers, the adoption of a cloud service is mentioned. A cloud server typically provides data storage and access to services through the Internet. In a dispenser, a cloud server can be used to store medication administration data, such as the time and amount of medication administered. In this way, the data can be easily accessed by the caregiver or the physician, who can check the patient's adherence to the therapy and make any changes to the prescription. For example, the use of a cloud server is central to the projects described in [18] and [37].

System Level By system level, we refer to components whose goal is the coordination and integration of other software and hardware components.

In [52] and [58], an MQTT (Message Queuing Telemetry Transport) service is used. MQTT is a lightweight and scalable messaging protocol used for communication between IoT devices. In a dispenser, an MQTT server can be used to send and receive data between the dispenser and other devices, such as an Android app or a cloud server. For example, the dispenser can send notification messages to the MQTT server regarding the time of drug administration, while the Android app can send commands to the dispenser to initiate the administration of a drug.

In [46], ROS (Robot Operating System) is used to control the mobile robot proposed as a dispenser. ROS is an open-source development framework used for creating robotic applications. It is used to manage sensors and actuators, such as proximity sensors to detect patient presence or actuators for drug administration.

4.3 RQ3: Which are the Interaction Modalities Provided by the Proposed Solutions?

To answer RQ3, we analyzed the solutions proposed in the selected papers with the purpose of identifying the most common ways provided to end users (patients, caregivers, and health professionals) to interact with the device itself or access the data collected during its use.

A first classification of interaction modalities may consider (i) the solutions that offer a direct interaction with the hardware of the dispenser for operating on it, receiving reminders, checking its status, or recharging it and (ii) the solutions that provide a software application, running on a separate device, as an additional or alternative way to interact with the pill dispenser.

4.3.1 Direct Interaction with the Pill Dispenser

Several hardware components and related managing software have been included in the proposed dispensers. We list them in the following by considering the input components first, then the output components, and finally those components that support both input and output (see a summary in Table 6).

Keypad and Buttons In 17 included papers, the use of a dedicated keypad or physical buttons as a method of interaction with the dispenser is described. The buttons are used to activate dispenser functions, such as pill dispensing or cancellation of warnings. Some examples can be found in [22, 24, 62] and [58]. A keypad can be available to set medication timings, as in [22, 26] and [30], thus providing a simple interface, usually combined with an LCD, to interact with the dispenser.

Microphone Only three solutions include a microphone. In [66, 71], we found that the microphone is used to make video calls with doctors, nurses, or caregivers, while only the dispenser described in [60], which is targeted to visually impaired people, exploits the microphone to help users take the right medicine at the right time. The interaction is however unnatural, since it consists of replaying system voice instructions obtained through pushing a button, and then the system recognizes users' sentences using Google voice kit.

Fingerprint Fingerprint authentication is a technology that allows the users to authenticate through the use of their fingerprints. In a dispenser, this technology could be used to ensure that only the authorized user can access the drugs contained within the dispenser. This technique is described in three papers: [20, 33] and [32].

Facial Recognition Facial recognition is an alternative way for user's authentication. For example, the user could register their face on the device, and the dispenser would use this registration to verify the user's identity each time it is opened. In [66], the use of facial recognition to access the device's system is mentioned.

Direct interaction componentTotalPapers				
Keypad and buttons	17	[22-24, 26, 28, 30, 35, 36, 38, 39, 54, 57, 58, 62, 64, 66, 70]		
Microphone	3	[60, 66, 71]		
Fingerprint	3	[20, 32, 33]		
Facial recognition	1	[66]		
LED	24	[15, 18, 21, 26, 27, 29 - 31, 33 - 36, 43 - 45, 51, 52, 55, 57, 58, 60 - 62, 66]		
LCD	17	[22-30, 32, 34, 39, 41, 43, 46, 54, 56]		
Speakers	13	[18, 21, 25, 28, 33, 42, 49, 57, 60–62, 64, 66]		
Buzzer	19	[22-24, 26, 27, 30, 32-36, 41-45, 52, 55, 56]		
Touchscreen	8	[29, 47, 50, 55, 64, 66, 68, 70]		

 Table 6
 Summary for Section 4.3.1

LED As to the output, the use of LEDs is the simplest and most intuitive way to signal something to the user. They are also often used because they are easy to install during the construction of a prototype. LEDs are used to indicate the status of the dispenser, such as whether the pills have been dispensed correctly or whether the dispenser needs to be charged. LEDs are mentioned in 24 papers, such as [18, 45] and [31].

LCD The LCD screen is used to display important information, such as the time the pills were dispensed, the number of pills remaining, the name of the medication, and the doses. In 17 included papers, the use of LCD is described, for example, in [29, 41] and [32].

Speakers Speakers are another widely used way of providing dispenser output. They can be used to provide voice information, such as instructions for use, pill dispensing notifications, or low battery alerts. They also enable text-to-speech for visually impaired users, as in [60] and [33]. The solutions that include a speaker are presented in 13 selected papers, e.g., [28, 57] and [61].

Buzzer The use of a buzzer is also widely used as a method to attract user's attention, as it is very common in the current use of smartphones. The vibration can be used as a tactile but also as an auditory signal to alert the user of the need to take pills or for other notifications. In 19 papers, the use of buzzers is described, for example, in [27, 34] and [36].

Touchscreen A dispenser endowed with a touchscreen provides the user with an easy and intuitive way both to obtain output information and to enter input in the system. It allows the user to select options, view information, and program the dispenser, according to doctor's prescriptions. This type of interaction can be found in eight analyzed papers, for example, in [29, 70] and [64].

4.3.2 Interaction Through Separate Devices

As previously mentioned, several papers propose the use of a separate device, with a dedicated software application, to interact with the pill dispenser. The separate device can be a smartphone, a personal computer, a robot, or a wearable device (see a summary in Table 7).

Device interaction component	Total	Papers
Smartphone	28	[15, 16, 18–20, 26, 33–42, 44, 46, 48, 49, 51–53, 56–59, 63]
Personal computer	2	[31, 61]
Mobile robot	3	[46, 48, 49]
Wearable device	1	[55]

Table 7Summary for Section 4.3.2

Smartphone The use of mobile applications or web applications running on smartphones is frequent in the selected papers (24 out of 56). They can represent the unique way to interact with the dispenser or can complement direct interaction with it. In the former case, the app covers all the offered functionalities, e.g., setting medication administration times, receiving notifications, and monitoring medication status, and it is used as well to activate the device, open/close its containers, and manage recharge. In the latter case, some of the functionalities are available through direct interaction with the dispenser (e.g., opening/closing containers). The implementation of mobile/web applications is described for example in [59, 63] and [51]. Notifications arrive as in-app notifications, informing the patient when it is time to take pills or if the dispenser needs maintenance, or alerting the caregiver/health professional if the patient has not taken the pills. In the solutions presented in [20, 26, 53] and [19], instead, notifications are sent to the user's smartphone via SMS, without the use of any ad hoc mobile applications.

Personal Computer In the solutions proposed in two papers, [31, 61], the interaction with the dispenser is mediated by the use of a personal computer where a specific software application runs and controls the device, possibly collecting all data about medication adherence and sending messages to the caregivers by email. The same software application can also be used to set medication doses and time schedule.

Mobile Robot Three papers developed pill dispensers as mobile robots ([46, 48, 49]). For instance, [46] created a robot able to map the indoor environment and bring medicines to the patients; the goal was removing the need for human assistance especially during the COVID-19 pandemic. All the three solutions presented in the papers above combine the use of the mobile robot with a mobile application to set medication timings and control the robot.

Wearable Device A wearable technology refers to a device that can be worn on the body, such as a bracelet or a watch. These devices are designed to collect and monitor data on the user's physical activities, health, and daily habits. The wearable device can monitor medication data and send notifications to the user when it is time to take pills. In addition, the collected data can be sent to a caregiver or healthcare professional for remote monitoring. For example, paper [55] describes a wearable device that enables patients to receive reminders to take their medication even when they are away from the main unit. This is achieved by synchronizing the pill dispensing schedule from the main unit (master) to the portable device (slave). When it is time to take the medication, the portable device will emit an audible beep, vibrate, and flash an LED light.

4.4 RQ4: What Kind of User Involvement Is Envisaged in the Proposed Solution?

To answer RQ4, we examined the solutions proposed in the selected papers considering whether they describe how the dispenser is recharged and how the medication therapy is managed. In particular, we focused on the role of users (patients, caregivers, doctors,

nurses, and pharmacists) in these two important activities. Only a subset of selected papers is herewith considered, paying attention to those ones presenting the most interesting answers to RQ4.

4.4.1 Pill Recharge Management

Recharging a dispenser is a challenging task, since it must be performed following doctor's prescriptions, avoiding contamination as much as possible, and managing the device component devoted to host the pills in an easy and safe way.

The different approaches to dispenser recharge are summarized in Table 8, even though several papers selected for this review do not discuss this problem at all.

Most of the papers mention that the recharge unit must be filled with medications manually by the user, who might be the patient or their caregiver, e.g., [23, 25, 27, 29, 36, 41, 45, 60, 62]. For instance, in [27], it is explicitly stated that the dispenser must be preloaded by the patient or their caregiver once a day; the box where pills must be placed consists of several compartments, each one devoted to a pill to be taken at a definite time of the day. In one of the most dated proposals, described in [23], a caregiver must put pills in the system canisters, placing each set of drugs (up to 3) in a separate canister; the system also encompasses a pump to fill a glass with water from a water tank (that must be recharged manually as well), which will be drunk by the patient to take the pills; this latter feature is not foreseen in any other proposal. In the recent solution proposed in [62], medicines must be put in plastic pillboxes; it is possible to use up to three pillboxes per day, i.e., one for the morning, one for noon, and one for the night. Then, the pillboxes must be put in the device that has three columns corresponding to morning, noon, and night. When it is time to take the medicines, the user must press a device button, then the pillbox comes out; after the medicine is taken, the pillbox is re-inserted again in the dispenser.

A slightly more sophisticated solution, still requiring manual recharging of the pill dispenser, is that presented in [44]: here, a caregiver is required to use a mobile app to set up the therapy for the patient by associating each pill type and the number of pills to a different pill box forming the recharge unit of the pill dispenser; in fact, this solution offers the possibility to dynamically change the number of pill boxes according to the patient's therapy; thus, once the pillboxes are physically set up and the therapy has been defined through the app, the last step is putting the pills in the boxes at regular times (e.g., every day, every week). A similar solution is implemented in [38], where

Pill recharge methodology	Total	Papers	
Manually by patient/caregiver	9	[23, 25, 27, 29, 36, 41, 45, 60, 62]	
Manually, supported by software application	2	[38, 44]	
Manually by healthcare professionals	1	[21]	
Prepackaged medication bags	3	[64, 68, 70]	

 Table 8
 Summary for Section 4.4.1

a manual refill of three containers (corresponding to morning, afternoon, and night) can be carried out with the guide of a web application.

The boxes of the dispenser presented in [21] must be refilled by certified pharmacists following prescriptions and timings defined by doctors; this is considered an aspect of the prototype to be improved, by developing a proper software application to reduce the complexity of the refilling process.

A more automatic and controlled approach is that proposed for example in [68]: in this solution, medications are packaged in multidose pouches by pharmacies and delivered to patients weekly or every 2 weeks; these pouches are then loaded in the medication dispenser to be dispensed at prescribed times, according to the patient's therapy. A similar solution is adopted in [70]: unit dose sachets of medicines for 1 to 4 weeks are machine-packed and delivered to nurses or delegated home care staff, who are in charge of bringing them to the patient's home and inserting them into the dispenser, along with entering the patient's medication information in the dedicated software application. As to the robotic medication-dispensing system discussed in [64], nurses are in charge of refilling it with medication bags every 2 weeks.

4.4.2 Therapy Management

In the dispensers presented in literature, different kinds of users are involved in therapy setting within the device, monitoring of therapy adherence, and data analysis of medication intakes. This led to develop different kinds of approaches through which the users set therapy or access data (see a summary in Table 9).

A web application is adopted in the solutions proposed in [38, 41, 45, 53]. For example, a simple web application, hosted by the microcontroller of the smart device, has been developed in [53] to help the user configure the pill dispenser entering the prescription details and other device settings (time, phone number, Wi-Fi network, etc.). Within the solution proposed in [38], the medical staff uses a web application for setting the timing and number of medicines to be taken by the patients; a dashboard is also available to allow monitoring medicine intake and other physiological parameters. The web application developed in [41] is a cloud-based solution, and its database can be accessed both by patients and doctors for monitoring the intake of pills; the doctors are also in charge of inserting medication doses in the cloud database that is directly connected with the controller of the dispenser and allows its correct operation.

Therapy management methodology	Total	Papers	
Web application	4	[38, 41, 45, 53]	
Mobile application	7	[35, 37, 42, 57–59, 63]	
Cross-platform mobile application	1	[57]	
Direct interaction	7	[20, 21, 23, 24, 26, 29, 36]	
Personal computer	1	[61]	
Prescription scan	2	[15, 32]	

 Table 9
 Summary for Section 4.4.2

Several solutions propose the use of a mobile application for therapy management (e.g., [35, 37, 42, 57–59, 63]). An Android mobile application has been developed in [35], which allows one healthcare professional to manage different patients with related pill schedules and receive notifications about medication intake; this mobile application uses Firebase as an intermediate database that the dispenser can access; the interaction with the dispenser by the patient thus makes data changes occur in Firebase and consequent update of the app status. Similarly, paper [59] describes an Android mobile application that provides patients with features for controlling the dispenser and managing the patient schedule and data. After the user logs in the app, this one displays the list of pills to be taken in a given day; when setting the therapy, the user will add pills and timings, as well as related alarms, and choose in what container of the connected dispenser the pills must be placed. With the app, the user can also manage dispenser refill, since the app provides information about the remaining pills. To take their pills, the user positions the smartphone with the app close to the dispenser for connection via Bluetooth; then, the smartphone sends signals to the dispenser about which container should be activated. Caregivers have a passive role in this solution: their phone numbers can be entered into the app, and they will be notified by SMS when patients do not take pills. An Android-based application is proposed also in [58]: it communicates with the dispenser through Wi-Fi and can be used by the patient to set medicine schedule and related alarms and monitor medicines available in the dispenser; the caregiver receives notifications by email in case the patient has not taken their pills.

The authors of [57] developed a cross-platform mobile application that provides two different user profiles: one for the elderly patient and the other for the caregiver. The patient can use the mobile app to control the pill scheduling and receive notifications when pills must be taken; in some cases, the patient is capable and allowed to change pill scheduling and recharge the dispenser with the help of the app. Using the app with the proper user profile, caregivers can manage pill scheduling, temporarily stop the dispensing of one or more pills, and receive notifications in case of non-adherence or when the dispenser must be refilled.

More rarely in the selected papers, the user must interact directly with the dispenser, using its touchscreen, touchpad, or other special buttons, to set up the therapy. The latter solution is adopted in [20, 21, 23, 24, 26, 29, 36]. For example, the dispenser proposed in [29] is endowed with touchscreen and a graphical user interface (GUI) that allow users to set medication timings and view important information. A mechanism based on interrupts is implemented: guided by the GUI, the user can set up to five interrupts that control the system time and send notifications for pill intake. The dispenser presented in [26], instead, exploits an LCD display and a keypad to set the timings of medications. Similarly, the solution proposed in [24] has an interface consisting of an LCD and a keyboard that support the user in entering medication timings and selecting elements from a preprogrammed menu; the dispenser can be programmed for up to four time moments per day, for 7 days.

A dedicated personal computer, connected to the dispenser, is the solution foreseen in [61], where a software application implements most of the functions to monitor the patients and interact with them through prerecorded commands. In particular, a medical adherence controller takes the medication therapy from a database and data from the storage unit sensors, generates control signals for the hardware components of the dispensers, sends messages to the caregiver by email, and updates the database with information about medication adherence. In this way, the application also supports caregivers to manage medications and review patients' activities.

There are finally specific solutions, such as that proposed in [32], in which the dispenser is able to scan the prescriptions and store in a MySQL database all information about medicines and their dispensing timings. A similar approach, based on scanning prescription data, is adopted in [15].

5 Discussion

In this section, we first summarize the findings emerged from the study (see Fig. 3), outlining the research gaps still present in the literature; we then discuss open issues and challenges to be considered while designing a novel solution in the field; we finally delineate implications for future research and describe the limitations of the research.

5.1 Findings

Through our RQ1, we discussed the development stage of the proposals described in the selected papers, ranging from design studies and proofs-of-concept to working prototypes and comprehensive solutions that have been deployed and evaluated. It emerged that few papers present a user evaluation of working prototypes, while even fewer papers discuss field studies of complete solutions: this highlights how the research on automated dispensers is often focused on hardware and/or software technology, and somehow overlooks the impact on the various stakeholders (patients,



Fig. 3 Classification of findings along RQ1-RQ4

caregivers, doctors, and the like), which is however crucial to achieve technology acceptance and relevant benefits at large. The few papers that describe some kind of system evaluation present user studies that involved a limited number of participants, usually only patients, while other stakeholders were more rarely involved. Among the eight papers that present studies about commercial devices (Section 4.1.3), only [64] adopts a comprehensive view on the problem by performing a long-term ethnographic study to investigate the impact of the technology on the everyday life of elderly people and their professional caregivers, while [70] collected qualitative feedback related to organizational complexity, and the other six papers limited their scope on testing the functionality and/or usability of specific devices. Furthermore, most of the papers do not describe a design methodology or architectural guidelines defining the roles and activities of the various stakeholders and the technical components that must be present in a comprehensive solution, thus limiting the chances for generalization and application in different contexts. Summing up, while there are many promising efforts, there are still significant gaps with respect to the development of mature solutions characterized by fully satisfactory user experience.

As to RQ2, findings were not particularly surprising. Papers describe the hardware and software components of their proposed prototypes, explaining their different features. Most of prototypes are based on flexible and low-cost microcontrollers like Arduino and Raspberry Pi and provide different levels of communication: from just sending text messages through a GSM module, to direct control of the dispenser through an application connected via Bluetooth, until remote access to online services and other IoT devices thanks to a Wi-Fi component. The latter is usually combined with a cloud-based software solution that contributes to create an articulated architecture, where the different stakeholders can access data according to their preferences and needs. This is crucial to create a comprehensive solution not focused only on pill dispensing. Therefore, the design and development of companion mobile or web applications become fundamental to increase the usability and acceptability of the dispenser for the different stakeholders. However, this type of architecture creates some dependencies, whose potential criticality is not discussed in the selected papers. In fact, dispenser operation will depend on the availability of external resources like Wi-Fi access and/or external software applications (at any level from smartphones to cloud services). Similarly, the integration of the dispenser in a richer IoT ecosystem, including sensors and wearable devices, may increase the number of features provided by the dispenser, but, at the same time, will add complexity to its management, including the programming of the behaviors of the IoT ecosystem, which may become a challenging task for patients, caregivers or other healthcare professionals. The endowment of dispensers with sensors for facial or finger recognition can be found in few proposed prototypes, even though this would significantly contribute to privacy and security. In a similar way, acquiring data about actual medication intake by elderly patients with cognitive impairments is a complex feature that is dealt with in a couple of selected papers, which provide however approaches aimed mainly at testing specific technologies devoted to this problem, rather than at integrating them within a complete solution to be deployed in real contexts.

The analysis of papers through the lens of RQ3 allowed us to identify different types of interaction modalities, from a simple operation on the medication dispenser

to its configuration and control by means of a software application running on an external device. Surely, the latter solution offers more flexibility and scalability, even though it might increase the complexity of use: an elderly patient may find it more difficult to interact with the graphical interface of a mobile app to set the dispenser and read its notifications, than operating on the physical buttons, and being alerted by blinking LEDs and/or buzzer sounds. Natural language interaction via voice-based messages is proposed in very few papers, which consider dispenser accessibility for visually impaired users and in any case encompass text-to-speech technology or registered messages, while natural language understanding is not yet proposed for this type of system. None of the selected papers mention some tailoring possibility that concerns interaction modalities, to accommodate different patients' attitudes and types of impairment (not only visual, but also hearing or cognitive). Papers that propose to adopt mobile robots to address the medication adherence problem are interesting, but at the moment, they are more suitable (and affordable) to healthcare structures than to use at home.

Through RQ4, we derived that there is a large spectrum of use contexts characterized by the involvement of different stakeholders in dispenser recharge and therapy management. As to dispenser preparation, it can be performed by the patient autonomously or by someone else close to them (i.e., an informal or formal caregiver), or preparation may be carried out in a separate site, like a pharmacy, by professional staff (e.g., pharmacists). Dispenser preparation is strictly related to the number of dispensers managed: in a home scenario, a single dispenser (or at most a small number of dispensers) is managed, while, in other scenarios (hospitals, retirement homes, and the like), a large number of dispensers needs to be managed together. Error prevention problems related to the latter situation are however not addressed in the selected papers. As to therapy management, the level of automation (and related cognitive effort) may vary a lot: since the dispenser is involved in a process that goes from therapy prescription, to planning medicine delivery and the relevant alerts, to the actual medicine intake, automated support can be present in all the process activities. For instance, prescription can be available in digital form and used to provide support for the correct dispenser filling and planning, or, alternatively, software applications (running on the dispenser or on a separate device) must be used for therapy definition; alerts can be used to remind of the moments when intake must occur, and some verification mechanism can be used to check that correct intake has occurred. While all these activities can benefit from automated support, in the selected papers, most of proposed solutions cover only some of them. Monitoring medication intake, if any, is supported in the majority of proposals through software applications that can receive notifications and provide information about patient's behavior, while very few solutions exploit cameras and image processing algorithms to verify actual medication intake. Finally, none of the papers mentions which corrective mechanisms are put in place if the patient forgets to take one or more pills at a given moment; for example, a temporary revision of the therapy plan could be activated automatically by the dispenser or the problem could be submitted to the caregiver or doctor, possibly with some decision support provided by the system.

5.2 Open Issues and Challenges

As evidenced previously, pill and medication dispensers represent a challenging subject, given their life-critical role, the variety of use contexts and stakeholders, and the possible involvement of fragile and/or impaired persons.

As such, they are a lively and largely open investigation subject: while many studies and solutions have been presented in the literature, several important aspects appear to have received limited attention and require further developments, as listed below:

- *Scalability*: Small scale and in particular individual use scenarios are more typically considered as they represent the simplest use case and testbed for solutions which are still in their infancy. However, scalability issues need to be explicitly addressed in several respects:
 - Possible adoption of the technology by mid- and large-size institutions like hospitals and residential houses
 - In case some part of the process is not carried out at individual level, e.g., if many individual dispensers are prepared at pharmacies
 - In case some external resources (e.g., cloud services) are shared by a potentially large number of individual dispensers

All the aspects in the design of a solution should undergo a scalability assessment. We remark in particular that if the phase of preparation and filling of the dispenser involves manual activities, they may represent a significant obstacle to scalability, as being slow and error-prone.

- *Integration*: The components and systems involved in the proposed prototypes, such as sensors, medication tracking software, and dispensing mechanism, are often developed independently and lack seamless integration. This lack of integration can lead to compatibility issues and limited functionality. It is essential for developers and manufacturers to focus on integrating these technologies effectively, ensuring smooth communication and interoperability among the different components. By achieving integration, they will be able to offer a better user experience, leading not only to improved medication management and patient care, but also to a more dynamic and open market for these solutions.
- Authentication and security: These aspects are not crucial during prototyping, but become critical in any actual deployment of the technology particularly in large-scale adoption scenarios. On the one hand, only authorized users should be allowed to interact with the dispenser. This is clearly crucial in contexts like hospitals, but is nevertheless important in domestic environments, e.g., in the presence of children. As obvious as this requirement is, it poses a challenge, since common authentication mechanisms may be difficult to use for elderly persons. On the other hand, especially if the dispenser is connected to the network, cybersecurity issues should be considered. Any attack potentially able to disrupt the dispenser operation may have life-critical consequences; appropriate prevention and mitigation measures should therefore be devised and put in place.
- *Dependability and safety*: A medication dispenser can be a life-critical device for some patients; hence, once the technology will be more mature, reliability

and fault-tolerance issues will need to be the subject of suitable analysis, using consolidated engineering techniques [72] and possibly leading to consider some redundancy in the dispenser design. The ability of the dispenser to work in the absence of other components or services (e.g., in the absence of network connections) and the possibility of easy reconfiguration (e.g., of Wi-Fi settings) should also be considered. This is particularly relevant in the case where the dispenser must be moved from one place to another (e.g., when the patient needs to move for any reason including vacation, work, hospitalization). On the user's side, safe usage considerations will need to be integrated in the interaction design, where the goal of providing a satisfactory user experience has to be balanced with the one of avoiding any unsafe operation.

• User experience and personalization: Physical and/or cognitive impairments of elderly patients are often underestimated, leading to the design of the interaction with pill dispensers based on graphical user interfaces (typically used in desktop, mobile, or web applications) which are unsuitable to patients' special needs. With the growing diffusion of advanced paradigms, such as speech-based and gesture-based interaction, more appropriate ways to interact with dispensers could be studied. This opens up the challenge of designing different interaction modalities for the same device, so that the users could adopt that one they prefer and consider most suitable to their capabilities. This could also give rise to the need of implementing functionalities that allows caregivers or professional staff to easily tailor the user interface according to the patients' characteristics and use context. Figure 4 presents a visual synthesis of open issues and challenges.

5.3 Implications for Future Research

In addition to the specific challenges discussed in the previous section, we envisage two general indications for future work, which we regard as long-term research goals.

5.3.1 Flexible Design with Modular Architecture

The variety of possible use scenarios suggests that hardly a "one size fits all" approach is suitable for the design of pill and medication dispensers: the features that make a solution perfectly fit for a use case may be disadvantageous and prevent acceptance in another case. Patients with different degrees of autonomy, the variety of possible roles and competencies of caregivers, and the optional involvement of professionals like pharmacists and doctors are just some examples of the features that give rise to an extremely heterogeneous landscape in terms of user needs and hence of system requirements. Developing a babel of different products, each addressing a specific niche of use, does not seem a viable approach either.

We suggest therefore that, in order to enable the adoption of dispensers at large, a modular and context-adaptable approach needs to be investigated.

As a first step, one might undertake the identification of a set of basic functions common to any use scenario, e.g., providing scheduled reminders and dispensing pills and medications accordingly. These functions should be implemented by a core device,



Fig. 4 Summaries of open issues and challenges

whose basic operation should be autonomous and robust, in order to ensure service continuity and also in case of failure of other components.

Then, a set of optional hardware or software components that can be added to the core device to enhance its capabilities should be designed in a modular fashion. To give a few examples, additional hardware components like a fingerprint reader might be devoted to user authentication while software components may extend the device functions (e.g., with automated therapy management) or support articulated scenarios with multiple roles (e.g., dispenser preparation by a pharmacist or therapy monitoring by a doctor).

To enable such a scenario, a suitable design methodology, oriented to openness and based on well-defined interfaces between components, should be adopted. In particular, we suggest that the functions of the core device should be associated with a standard set of application programming interfaces (APIs), whose definition represents a challenging task by itself. The broad adoption of such a standard would ensure, at least at a basic level, the interoperability between dispensers by different producers and device-independent software tools for dispenser configuration, control, monitoring, and so on.

While there is undoubtedly a significant gap between the current state of the art and the scenario envisaged above, we believe that stimulating research in this direction is crucial to overcome some critical obstacles to the adoption of dispenser technology beyond small-scale applications. Moreover, it would represent an enabling factor with respect to the more ambitious long-term research goal described next.

5.3.2 Towards Therapy Management Systems

The major goal of pill and medication dispensers is ensuring therapy adherence. However, a patient therapy may include aspects that cannot be managed by a dispenser. For instance, the use of an aerosol machine at home may be part of a therapy plan, and this cannot be reduced to the intake of something provided at the right moment by the dispenser.

Thus, the study of smart dispensers, as challenging as it is by itself, can be regarded as just a first step towards the more challenging goal of developing a patient-centered therapy management system of which the dispenser will be an essential, but not unique, component. In fact, this would require the realization of an ecosystem of therapysupporting devices and of software tools for therapy planning, administering, and monitoring, along with the characterization of the various use contexts and of the roles of the involved people (mainly patients, caregivers, pharmacists, doctors).

Supporting technologies for such a scenario (in particular Internet of Things and cloud-based services) are available, while the main difficulty is to put together the pieces of the puzzle, whose complexity essentially lies in tailoring a set of technologies to the needs of the people they are intended to serve.

We believe that the development of human-centered projects, based on the use of dispensers, will represent a fruitful seed in this respect and can pave the way to the investigation of richer and more ambitious therapy management solutions.

5.4 Limitations of the Research

This systematic literature review has the limitations indicated in the following. First, we limited our research to the period January 1st, 2013–February 6th, 2023, on Scopus (about 10 years). As already mentioned, the decision about the publication period was based on the observation that technologies for pill and medication dispensers rapidly evolve, making solutions obsolete very soon. In addition, we found that the most interesting proposals from a human-centered perspective have been published in the last 4 years. Then, we have had to limit paper search to the beginning of February 2023 to start paper analysis. At the date of submission, further papers addressing the topic of this study could have been published. However, as already mentioned, a further search on PubMed on October 23rd, 2023, did not find additional relevant work.

Second, our query string was based on the terms "Pill dispenser" and "Medication dispenser," following the idea of the survey [12]. While we believe that these terms provide a very good coverage of the topic we aimed to address, we cannot exclude that in some other relevant papers, further and possibly more specific terms were used.

Third, our overarching research question only considered literature work; thus, we did examine only indirectly the existing commercial solutions available on the market, whose evaluation is reported in some of the selected papers (see Section 4.1.3). While other surveys on mobile health apps [73] have been successfully conducted

resorting to app descriptions available online, in our case, analyzing from humancentered perspective commercial solutions not described in scientific publications would have required specific user studies, which were beyond the limit of the present paper. Surveys of mobile apps like the ones reported in [11, 73] are complementary to our work.

Fourth, the focus of our review was investigating the issues and challenges that could emerge from a human-centered perspective, by paying attention to the most interesting socio-technical aspects related to the development and deployment of medication dispenser technology. Other aspects like for instance the impact on therapy adherence of medication dispensers were out of the scope of our analysis and can be found in other literature reviews as indicated in Section 2. One of the possible outcomes of a systematic review is defining a taxonomy for a given research field (e.g., [11, 73– [75]) which can be built from the findings in the reviewed papers following a suitable methodology, like the one proposed in [76]. Producing a taxonomy was not a goal of our work, which is focused on analyzing the human-centered perspective in the variety of proposed approaches to pill and medication dispensers. In particular, the requirement that the taxonomy considers characteristics that are common across all the articles [76] makes impractical the creation of a significant taxonomy in the context we considered, given the diversity of the surveyed papers. To address the difficulties arising from the requirement mentioned above, in [11], a comprehensive framework is proposed. In the same spirit, we provided a synthetic overview of our findings in Fig. 3.

6 Conclusion

This paper presented a systematic review on pill and medication dispensers from a human-centered perspective. To this purpose, 426 papers indexed in Scopus and PubMed have been scrutinized, finding 56 papers relevant for our overarching research question. These papers have been analyzed considering the stage of development of the proposed solution, the developed software and hardware technology, the type of user interaction provided by the solution, and the type of involvement of different stakeholders in the use and management of the proposed solution.

The findings show the existence of several gaps in the design, development, and deployment of comprehensive healthcare solutions that involve the use of pill and medication dispensers. The most important open issues and challenges have been derived from the analysis, highlighting how they concern solution scalability, system integration, authentication and security, dependability and safety, user experience, and personalization. As a final contribution of the review, two general indications for future work have been envisaged, suggesting the need for further research on design approaches to developing socio-technical solutions that may foster medication adherence in a broader sense, thus providing more comprehensive solutions for the improvement of patient health.

Appendix. PRISMA items considered in the study

Table 10 lists all the PRISMA items used in this systematic review, along with their descriptions (as in [5]) and their corresponding location in this paper.

Item	Description	Paper's section
1	Identify the report as a systematic review	Title of the paper
2	Abstract	Abstract
4	Provide an explicit statement of the objective(s) or question(s) the review addresses	Introduction (Section 1)
5	Specify the inclusion and exclusion criteria for the review and how studies were grouped for the syntheses	Methodology (Section 3)
6	Specify all databases, registers, web- sites, organizations, reference lists and other sources searched or consulted to identify studies. Specify the date when each source was last searched or con- sulted	Methodology (Section 3)
7	Present the full search strategies for all databases, registers and websites, including any filters and limits used	Methodology (Section 3)
8	Specify the methods used to decide whether a study met the inclusion criteria of the review, including how many reviewers screened each record and each report retrieved, whether they worked independently, and if applicable, details of automation tools used in the process	Methodology (Section 3)
9	Specify the methods used to col- lect data from reports, including how many reviewers collected data from each report, whether they worked independently, any processes for obtaining or confirming data from study investigators, and if applicable, details of automation tools used in the process	Methodology (Section 3)
10a	List and define all outcomes for which data were sought. Specify whether all results that were compatible with each outcome domain in each study were sought (e.g., for all measures, time points, analyses), and if not, the meth- ods used to decide which results to collect	Methodology (Section 3)

 Table 10
 The description and position in this paper of the PRISMA items considered in the study are provided

Item	Description	Paper's section	
16a	Describe the results of the search and selection process, from the number of records identified in the search to the number of studies included in the review, ideally using a flow diagram	Results (Section 4)	
17	Cite each included study and present its characteristics	Results (Section 4)	
23a	Provide a general interpretation of the results in the context of other evidence	Discussion (Section 5)	
23d	Discuss implications of the results for practice, policy, and future research	Discussion (Section 5)	

Table 10 continued

Author Contribution All authors contributed to the literature review conception and design. The literature search and data analysis were performed by LG and DF. All the authors contributed equally to the writing of the manuscript.

Funding Open access funding provided by Università degli Studi di Brescia within the CRUI-CARE Agreement. The PhD scholarship of Luigi Gargioni is co-funded by the Italian Ministry, Piano Nazionale di Ripresa e Resilienza (PNRR) and Antares Vision S.p.A.

Availability of Data and Material The dataset of the papers analyzed for this manuscript is available from the corresponding author on request.

Code Availability Not applicable

Declarations

Ethics Approval Not applicable

Consent to Participate Not applicable

Conflict of Interest The authors declare no competing interests.

Human Ethics Not applicable

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Eurostat: Ageing Europe Statistics on population developments (2020). https://ec.europa.eu/eurostat/ statistics-explained/index.php?title=Ageing_Europe_-_statistics_on_population_developments. Accessed 17 May 2023
- Wilhelmsen NC, Eriksson T (2019) Medication adherence interventions and outcomes: an overview of systematic reviews. Eur J Hosp Pharm 26(4):187–192
- 3. Organization WH et al (2003) Adherence to long-term therapies: evidence for action. World Health Organization
- 4. Aseeri M, Banasser G, Baduhduh O, Baksh S, Ghalibi N (2020) Evaluation of medication error incident reports at a tertiary care hospital. Pharmacy 8(2):69
- Page MJ, McKenzie JE, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, Shamseer L, Tetzlaff JM, Akl EA, Brennan SE, Chou R, Glanville J, Grimshaw JM, Hróbjartsson A, Lalu MM, Li T, Loder EW, Mayo-Wilson E, McDonald S, McGuinness LA, Stewart LA, Thomas J, Tricco AC, Welch VA, Whiting P, Moher D (2021) The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ 372. https://doi.org/10.1136/bmj.n71
- Kumar SK, Manimegalai R, Rajeswari A, Deekshita R, Dhineshkumar M, Manikandan G (2021) A literature review: performance evaluation of wearable system with pill dispenser box for post COVID elderly patients. In: 2021 3rd International conference on advances in computing, communication control and networking (ICAC3N), pp 2008–2014. https://doi.org/10.1109/ICAC3N53548.2021.9725733
- Sriram V, Jenkinson C, Peters M (2019) Informal carers' experience of assistive technology use in dementia care at home: a systematic review. BMC Ger 19
- 8. Sriram V, Jenkinson C, Peters M (2019) Informal carers' experience and outcomes of assistive technology use in dementia care in the community: a systematic review protocol. Syst Rev 8
- Faisal S, Ivo J, Lee C, Carter C, Patel T (2022) The usability, acceptability, and functionality of smart oral multidose dispensing systems for medication adherence: a scoping review. J Pharm Pract 35(3):455–468. https://doi.org/10.1177/0897190020977756
- Steinkamp JM, Goldblatt N, Borodovsky JT, LaVertu A, Kronish IM, Marsch LA, Schuman-Olivier Z (2019) Technological interventions for medication adherence in adult mental health and substance use disorders: a systematic review. JMIR Ment Health 6(3):12493. https://doi.org/10.2196/12493
- 11. Singh N, Varshney U (2020) IT-based reminders for medication adherence: systematic review, taxonomy, framework and research directions. Eur J Inf Syst 29(1):84–108
- Pinto, JF, Vilaça JL, Dias NS (2021) A review of current pill organizers and dispensers. In: 2021 IEEE 9th International conference on serious games and applications for health (SeGAH), pp 1–8. https:// doi.org/10.1109/SEGAH52098.2021.9551894
- 13. Bar-Ilan J (2018) Tale of three databases: the implication of coverage demonstrated for a sample query. Front Res Metr Anal 3:6
- Cohen J (1960) A coefficient of agreement for nominal scales. Educ Psychol Meas 20(1):37–46. https:// doi.org/10.1177/001316446002000104
- Cakaloglu T (2017) Medi-Deep: deep control in a medication usage. In: 2017 IEEE International conference on bioinformatics and biomedicine (BIBM), IEEE, pp 899–904
- Urzaiz G, Murillo E, Arjona S, Hervas R, Fontecha J, Bravo J (2013) An integral medicine taking solution for mild and moderate Alzheimer patients. In: Ambient Assisted Living and Active aging: 5th International Work-Conference, IWAAL 2013, Carrillo, Costa Rica, December 2-6, 2013, Proceedings 5, Springer, pp 104–111
- Reichelt F, Schmid P, Maier T, Sahlab N, Jazdi-Motlagh N, Weyrich M, Meyer-Philippi G, Kalka G, Matschke M (2020) User centered design of a pill dispenser for the elderly. In: Human systems engineering and design II: Proceedings of the 2nd International Conference on human systems engineering and design (IHSED2019): Future Trends and Applications, September 16-18, 2019, Universität der Bundeswehr München, Munich, Germany, Springer, pp 699–704
- 18. Sahlab N, Jazdi N, Weyrich M, Schmid P, Reichelt F, Maier T, Meyer-Philippi G, Matschke M, Kalka G (2020) Development of an intelligent pill dispenser based on an IoT-approach. In: Human systems engineering and design II: Proceedings of the 2nd International conference on human systems engineering and design (IHSED2019): Future trends and applications, September 16-18, 2019, Universität der Bundeswehr München, Munich, Germany, Springer, pp 33–39
- Hussein Shukur M (2019) Design a mobile medication dispenser based on IoT technology. Int J Innov Creativity Chang 6(2):242–250

- Patil C, Lightwala N, Sherdiwala M, Vibhute AD, Naik SA, Mali SM (2022) An IoT based smart medicine dispenser model for healthcare. In: 2022 IEEE World conference on applied intelligence and computing (AIC), IEEE, pp 391–395
- Pandey PS, Raghuwanshi SK, Tomar GS (2018) The real time hardware of smart medicine dispenser to reduce the adverse drugs reactions. In: 2018 International conference on advances in computing and communication engineering (ICACCE), IEEE, pp 413–418
- MohanaPriya D, Deepika V, Priya MS, Yogeswari CS (2020) A real time support system to impart medicine using smart dispenser. In: 2020 International conference on system, computation, automation and networking (ICSCAN), IEEE, pp 1–10
- Mohammed MNA, Ahmed AHM, Salih TM (2013) Designing low cost digital dose reminder system. In: 2013 international conference on computing, electrical and electronic engineering (ICCEEE), IEEE, pp 357–362
- Fărcaş C, Ciocan I, Palaghiţă N, Fizeşan R (2015) Weekly electronic pills dispenser with circular containers. In: 2015 IEEE 21st International symposium for design and technology in electronic packaging (SIITME), IEEE, pp 125–129
- Chandana N, Kumar M, Hegde GS, Raj KS et al (2022) IoT based medicine reminder and dispensing machine. In: 2022 Second international conference on artificial intelligence and smart energy (ICAIS), IEEE, pp 1060–1066
- Jagadeeshwaran A, Kumar HS, Sayeed SS, Vaithilingam CA (2021) Smart pill reminder. In: 2021 First International conference on advances in computing and future communication technologies (ICACFCT), IEEE, pp 174–178
- Jabeena, A., Sahu, A.K., Roy, R., Basha, N.S (2017) Automatic pill reminder for easy supervision. In: 2017 International Conference on Intelligent Sustainable Systems (ICISS), IEEE, pp 630–637
- Mugisha GA, Uzoka F-M, Nwafor-Okoli C (2017) A framework for low cost automatic pill dispensing unit for medication management. In: 2017 IST-Africa Week Conference (IST-Africa), IEEE, pp 1–10
- Chariah C, Kumia N, Zubarriain JR, He S, Rahemi H, Jack H (2016) Maker: automated pill dispenser. In: 2016 ASEE Annual conference & exposition
- Ibitoye MO, Raji AO, Nafiu SO (2019) Inexpensive automated medication dispenser for persons with neurodegenerative illnesses in low resource settings. J Med Eng Technol 43(8):451–456
- Caliolio MH, Frias J, Paglinawan C (2022) Integration of a smart medicine container with medicine sorting using YOLOv4 and OCR. In: 2022 13th International conference on computing communication and networking technologies (ICCCNT), IEEE, pp 1–6
- 32. Rumi RI, Pavel MI, Islam E, Shakir MB, Hossain MA (2019) IoT enabled prescription reading smart medicine dispenser implementing maximally stable extremal regions and OCR. In: 2019 Third international conference on I-SMAC (IoT in social, mobile, analytics and cloud)(I-SMAC), IEEE, pp 134–138
- YR M, Lohith N, SV B et al (2019) MEDIC-the SmartMedicine dispenser. In: Proceedings of the second international conference on emerging trends in science & technologies for engineering systems (ICETSE-2019)
- Madankar A, Agrawal A, Yede V (2021) IoT based advance pill reminder system for distinct patients. In: 2021 Fifth international conference on I-SMAC (IoT in social, mobile, analytics and cloud)(I-SMAC), IEEE, pp 161–165
- Rosdi W, Suhaimi S, Lazam NM, Alias A, Abdullah F, Azemi S (2021) Smart pill dispenser with monitoring system. In: 2021 IEEE symposium on wireless technology & applications (ISWTA), IEEE, pp 58–62
- Meghla RT, Deowan ME, Nuhel AK, Sazid MM, Ekbal MN, Mahamud MH (2022) An Internet of Things (IoT)-based smart automatic medication dispenser with an integrated web application for patient diagnosis. In: 2022 5th International conference of computer and informatics engineering (IC2IE), IEEE, pp 16–21
- Bharadwaj SA, Yarravarapu D, Reddy SCK, Prudhvi T, Sandeep K, Reddy OSD (2017) Enhancing healthcare using m-care box (monitoring non-compliance of medication). In: 2017 International Conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud)(I-SMAC), IEEE, pp 352–356
- Umamaheswari S, Babu MR, Prasanth DD, Bharathi HN, Alagumeenaakshi M (2021) Smart medical dispenser with pulse meter. In: 2021 International conference on advancements in electrical, electronics, communication, computing and automation (ICAECA), IEEE, pp 1–5
- Ahmed A, Ruman MR, Barua A (2019) Home medication for elderly sick people. In: 2019 5th International conference on advances in electrical engineering (ICAEE), IEEE, pp 369–374

- Tsai K-L, Liau B-Y, Hung Y-M, Yu G-J, Wang Y-C (2020) Development of smart pillbox using 3D printing technology and convolutional neural network image recognition. Sens Mater 32(5):1907–1912
- 41. Ramkumar J, Karthikeyan C, Vamsidhar E, Dattatraya KN (2020) Automated pill dispenser application based on IoT for patient medication. IoT and ICT for Healthcare Applications, 231–253
- Swarnalatha E, Laxmi N (2022) IoT based digital pill developments for med-tech field advancements. In: AIP Conference Proceedings, AIP Publishing LLC, vol 2519, p 030079
- Paviya R, Prabakar S, Porkumaran K, Saman ABS (2022) Automated pill and syringe dispenser. In: International conference on artificial intelligence for smart community: AISC 2020, 17–18 December, Universiti Teknologi Petronas, Malaysia, Springer, pp 919–925
- 44. Buono P, Cassano F, Legretto A, Piccinno A (2018) A modular pill dispenser supporting therapies at home. In: Current trends in web engineering: ICWE 2018 International Workshops, MATWEP, EnWot, KD-WEB, WEOD, TourismKG, Cáceres, Spain, June 5, 2018, Revised Selected Papers 18, Springer, pp 71–82
- Mondragón A, De Hoyos A, Trejo A, González M, Ponce H (2017) Medi-Kit: developing a solution to improve attention on medical treatment. In: 2017 IEEE Mexican humanitarian technology conference (MHTC), IEEE, pp 28–33
- Ganga U, Jayashree S, Sakthivel V, Samiullah S, Harikumar M (2021) Indoor self-mapping pill dispensing device using slam algorithm. In: 2021 6th International conference on communication and electronics systems (ICCES), IEEE, pp 891–896
- Chelvam YK, Zamin N (2014) M3DITRACK3R: a design of an automated patient tracking and medicine dispensing mobile robot for senior citizens. In: 2014 International conference on computer, communications, and control technology (I4CT), IEEE, pp 36–41
- Karthikeyan K, Babu ED, Ranjith S, Arunkumar S (2021) Smart pill dispenser for aged patients. In: 2021 International conference on advancements in electrical, electronics, communication, computing and automation (ICAECA), IEEE, pp 1–5
- Bathirappan K, Pandurengan DB, Nawaz JY, Arulrajan JR (2022) Patient monitoring and medicine dispenser robot. In: AIP Conference Proceedings, AIP Publishing LLC, vol 2494, p 030004
- Harjumaa M, Idigoras I, Isomursu M, Garzo A (2014) Expectations and user experience of a multimodal medicine management system for older users. J Assist Technol 8(2):51–63
- 51. Barrué C, Vila R, Martinez AB, Martín M, Vázquez-Salceda J, Simó R, Barroso M, Cortés U (2019) DIALCAT: diabetes as an accelerator of cognitive impairment and Alzheimer's disease, comprehensive approach and adherence to treatment. Computación y Sistemas 23(2):351–363
- 52. Chavez E, Sifuentes B, Vidal R, Grados J, Rubiños S, Cuzcano A (2020) Remote monitoring applying IoT to improve control of medication adherence in geriatric patients with a complex treatment regimen, Lima-Peru. In: Proceedings of the 2020 3rd international conference on electronics and electrical engineering technology, pp 49–54
- Bandara C, Sandanayake AD, Kodithuwakku Y, Logeeshan V (2022) Automated medicinal-pill dispenser with cellular and Wi-Fi IoT integration. In: 2022 IEEE World AI IoT Congress (AIIoT), IEEE, pp 01–07
- Mugisha GA, Muhumuza C, Uzoka F-M, Nwafor-Okoli C, Nabunje J, Arindagye M, Bukenya JN (2021) Usability evaluation of low-cost smart pill dispenser by health care practitioners. In: Proceedings of the future technologies conference (FTC) 2020, Springer, Vol 3, pp 17–29
- 55. Duran B, Villegas D, Valencia S, Rahemi H, Benalla M, Bustamante M, Elzawawy A (2020) Autonomous medication distributor through implementation of Bresenham's line algorithm. In: Proc. LACCEI Int. Multi-conference Eng. Educ. Technol, pp 1–9
- Aziz SA, Sajak AAB, Ramly R, Zulfakar MH (2022) IoT automated pill dispenser for elderly care. In: Proceedings of seventh international congress on information and communication technology: ICICT 2022, London, Springer, vol 2 pp 13–27
- Casciaro S, Massa L, Sergi I, Patrono L (2020) A smart pill dispenser to support elderly people in medication adherence. In: 2020 5th International conference on smart and sustainable technologies (SpliTech), IEEE, pp 1–6
- Ayu WS, Septendhi R, Azi B, Hutabarat MT, Raharja YS (2018) Android application design of automatic tablet dispenser using MQTT protocol communication. In: 2018 International symposium on electronics and smart devices (ISESD), IEEE, pp 1–6
- Kassem A, Antoun W, Hamad M, El-Moucary C (2019) A comprehensive approach for a smart medication dispenser. Int J Comput Digit Syst 8(02):131–141

- Al-Haider AJ, Al-Sharshani SM, Al-Sheraim HS, Subramanian N, Al-Maadeed S, Chaari M (2020) Smart medicine planner for visually impaired people. In: 2020 IEEE International conference on informatics, IoT, and enabling technologies (ICIoT), IEEE, pp 361–366
- Moshnyaga V, Koyanagi M, Hirayama F, Takahama A, Hashimoto K (2016) A medication adherence monitoring system for people with dementia. In: 2016 IEEE International conference on systems, man, and cybernetics (SMC), IEEE, pp 000194–000199
- Rahman MM, Aktar R, Dey SK (2022) Design and implementation of a low-cost automated medicine dispenser. In: 2022 International conference on advancement in electrical and electronic engineering (ICAEEE), IEEE, pp 1–6
- Zijp TR, Touw DJ, Boven JF (2020) User acceptability and technical robustness evaluation of a novel smart pill bottle prototype designed to support medication adherence. Patient preference and adherence, 625–634
- Airola E, Rasi P (2020) Domestication of a robotic medication-dispensing service among older people in Finnish Lapland. Hum Technol 16(2):117
- 65. Lima O, Terroso M, Dias N, Vilaça JL, Matos D (2021) Development of the business model and user experience for a pill dispenser: a designer perspective. In: Advances in design and digital communication ii: proceedings of the 5th international conference on design and digital communication, Digicom 2021, November 4–6, 2021, Barcelos, Portugal, Springer, pp 552–564
- 66. Lima O, Terroso M, Dias N, Vilaça, JL, Matos D (2022) Development of a pill dispenser: system requirements and product architecture. In: Advances in design and digital communication iii: proceedings of the 6th international conference on design and digital communication, Digicom 2022, November 3–5, 2022, Barcelos, Portugal, Springer, pp 336–344
- 67. Patel T, Ivo J, Faisal S, McDougall A, Carducci J, Pritchard S, Chang F (2020) A prospective study of usability and workload of electronic medication adherence products by older adults, caregivers, and health care providers. J Med Internet Res 22(6):18073
- Patel T, Ivo J, Pitre T, Faisal S, Antunes K, Oda K et al (2022) An in-home medication dispensing system to support medication adherence for patients with chronic conditions in the community setting: prospective observational pilot study. JMIR Formative Res 6(5):34906
- Hannink K, Ter Brake L, Oonk N, Wertenbroek A, Piek M, Vree-Egberts L, Faber M, Palen J, Dorresteijn L (2019) A randomized controlled efficacy study of the Medido medication dispenser in Parkinson's disease. BMC Ger 19:1–8
- 70. Iqbal S, Jokela P, Hammar T, Nilsson A-L (2021) Sustainable healthcare systems. A holistic perspective on the use and impact of medication management robots in home healthcare. Scand J Inf Syst 33(2):6
- 71. Kumar SK, Manimegalai R, Rajeswari A, Deekshita R, Dhineshkumar M, Manikandan G (2021) A literature review: performance evaluation of wearable system with pill dispenser box for post COVID elderly patients. In: 2021 3rd International conference on advances in computing, communication control and networking (ICAC3N), IEEE, pp 2008–2014
- 72. Trivedi KS, Bobbio A (2017) Reliability and availability engineering. Cambridge University Press
- Varshney U, Singh N, Bourgeois AG, Dube SR (2021) Review, Assess, Classify, and Evaluate (RACE): a framework for studying m-health apps and its application for opioid apps. J Am Med Inform Assoc 29(3):520–535
- 74. Hors-Fraile S, Rivera-Romero O, Schneider F, Fernandez-Luque L, Luna-Perejon F, Civit-Balcells A, de Vries H (2018) Analyzing recommender systems for health promotion using a multidisciplinary taxonomy: a scoping review. Int J Med Inform 114:143–155
- Hernandez N, Castro L, Medina-Quero J, Favela J, Michan L, Mortenson WB (2021) Scoping review of healthcare literature on mobile, wearable, and textile sensing technology for continuous monitoring. J Healthc Inform Res 5(3):270–299
- Nickerson RC, Varshney U, Muntermann J (2013) A method for taxonomy development and its application in information systems. Eur J Inf Syst 22(3):336–359

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.